Author's response to reviews

Title: Analysis of the XRCC1 gene as a modifier of the cerebral response in ischemic stroke.

Authors:

Ana V Dutra (dutracacd@hotmail.com)
Hsiu-Fen Lin (sflin@kmu.edu.tw)
Suh-Hang H Juo (hjuo@c2b2.columbia.edu)
Harvey Mohrenweiser (hmohrenw@uci.edu)
Souvik Sen (SenS@glial.med.unc.edu)
Raji P Grewal (rgrewal@solarishs.org)

Version: 4 Date: 2 November 2006

Author's response to reviews: see over
November 2, 2006

The BioMed Central Editorial Production Team:

Journal: BMC Medical Genetics
MS : 5690798561073261

I have made the changes as requested:

1) Author list in the manuscript is exactly as in the manuscript
2) The country has been added to the affiliation details
3) The Introduction has been renamed Background

I have also checked for minor errors.

Sincerely:

Dr. Grewal